A New Technique for the Estimation of Cardiac Motion in Echocardiography Based on Transverse Oscillations: a preliminary evaluation *in silico* and a feasibility demonstration *in vivo*

Martino Alessandrini, Adrian Basarab, Loic Boussel, Xinxin Guo, André Serusclat, Denis Friboulet, Denis Kouamé, Olivier Bernard, Hervé Liebgott

Abstract-Quantification of regional myocardial motion and deformation from cardiac ultrasound has been fostering consid-2 erable research efforts. Despite the tremendous improvements 3 done in the field, all existing approaches still face a common limitation which is intrinsically connected with the formation 5 of the ultrasound images. Specifically, the reduced lateral res-6 olution and the absence of a phase information in the lateral direction highly limit the accuracy in the computation of lateral 8 displacements. In this context, this paper introduces a novel q 10 setup for the estimation of cardiac motion with ultrasound. The framework includes an unconventional beamforming technique 11 and a dedicated motion estimation algorithm. The beamformer 12 aims at introducing phase information in the lateral direction by 13 producing transverse oscillations. The estimator directly exploits 14 the phase information in the two directions by decomposing 15 the image into two 2D single-orthant analytic signals. The 16 17 displacement is then computed by assuming time conservation of the two associated image phases. A local affine displacement 18 model accounts for typical contraction/expansion, rotation and 19 shear of myocardial tissue. 20

The proposed framework was evaluated in silico on five ultra-21 realistic simulated echocardiographic sequences corresponding 22 to three parasternal short-axis and two apical four-chamber 23 acquisitions. The algorithm was contrasted against other two 24 phase-based solutions exploiting the presence of transverse oscil-25 lations and against block-matching on standard images without 26 transverse oscillations. The evaluation revealed that all algo-27 rithms exploiting transverse oscillations were able to estimate 28 lateral displacements with a better accuracy as compared to 29 block matching, leading to an overall higher precision in the 30 computation of the cardiac strain. Moreover, among the phase-31 based solutions considered, the proposed one was found to be in 32 average the more precise and reliable. 33

An implementation of the new beamforming strategy on a research ultrasound platform is also presented in this paper along with a preliminary *in vivo* evaluation on one healthy subject.

Index Terms—echocardiography, latreal displacements, motion
 estimation, radiofrequency signal, transverse oscillations, multi dimensional Hilbert transform, cardiac strain.

André Serusclat is with the Department of Radiology, Louis Pradel Hospital, Lyon, France.

I. INTRODUCTION

Cardiovascular diseases are the leading cause of deaths in 41 the world (48%), and it is projected that the annual number 42 of deaths due to cardiovascular disease will increase from 17 43 millions in 2008 to 25 millions in 2030 [1]. In this context, 44 clinical assessment of the cardiac function is essential for the 45 diagnosis and treatment of heart diseases. Among available 46 imaging techniques echocardiography has received special 47 attention, since it offers high temporal resolution while being 48 of relatively low cost. Moreover, cardiac motion estimation 49 and the derived strain measures performed from ultrasound 50 image sequences has proven to be a valuable tool for assessing 51 cardiac function [2]-[6]. As a consequence, the development 52 of motion estimation techniques from cardiac ultrasound data 53 has a long history, dating back to the late eighties [7], and is 54 still the topic of active research [8]–[12]. 55

Following [13], most common approaches can be grouped 56 in three main classes. A first family of methods is based on 57 the differential technique known as optical flow. The earlier 58 attempts towards automated cardiac motion estimation belong 59 to this class [7], [14], [15]. Since they rely on the local analysis 60 of spatial and temporal gradients, these methods may fail at 61 estimating large inter-frame cardiac motion. This implies using 62 multi-scale strategies or a block-matching initialization to 63 provide a reliable first-order estimate of the displacement [16]-64 [18]. A second family is referred to as speckle tracking, and 65 consists in finding the best match, as defined by the adopted 66 similarity measure, between two blocks extracted from two 67 subsequent frames. Most common similarity measures include 68 cross-correlation (CC) [19], [20], sum of absolute differences 69 (SAD) [21] or sum of squared differences (SSD) [22]. It 70 was shown in [23] that these measures provide the maximum 71 likelihood estimate of the displacement for a given statistical 72 distribution of the image noise (Laplacian for SAD, Gaussian 73 for SSD) and, following that observation, a new measure 74 based on a Rayleigh distributed multiplicative noise was there 75 introduced. Similar lines of reasoning have been exploited 76 in [24]-[26]. Finally, several authors proposed to estimate 77 cardiac motion by using non-rigid image registration, i.e. by 78 computing a global deformation map warping a given frame on 79 a reference one. The deformation field can be either discrete or 80 parametric and is generally computed by minimizing a given 81

Martino Alessandrini, Loic Boussel, Xinxin Guo, Denis Friboulet, Olivier Bernard and Hervé Liebgott are with Université de Lyon, CREATIS; CNRS UMR5220; Inserm U1044; INSA-Lyon; Universit Lyon 1, France. E-mail: liebgott@creatis.insa-lyon.fr

Adrian Basarab and Denis Kouamé are with IRIT; CNRS UMR5505, France. E-mail: adrian.basarab@irit.fr

cost function. In [27] the deformation field is represented on a
B-spline basis and estimated by applying a SSD-like similarity
measure to the image intensity. In [28] a discrete deformation
function based on intensity and phase information is used.
Since image registration is formulated as an inverse problem,
it allows easily introducing *a priori* such as smoothness [11],
[29], or incompressibility [8], [11].

Most of the approaches described above operate on conven-89 tional envelope-detected images, i.e. obtained through demod-90 ulation of the radio-frequency (RF) signal. Recently, several 91 studies have proposed performing speckle tracking by using 92 the RF signal instead. Since the RF signal contains high 93 frequencies it is indeed better adapted to the estimation of 94 small motions (typically on the order of the emitted pulse 95 wavelength). This is done by using either time-domain correla-96 97 tion or phase difference estimation [30]. This type of technique is particularly used for *cardiac elastography* and examples 98 include the work by Lubinski [31], Chen [32], D'hooge [33], 99 Lopata [34], [35] and Konofagou [36]. RF-based speckle 100 tracking is, however, currently not widespread in the field of 101 echocardiography because its high motion sensitivity implies 102 high frame rates [33], [37]. A comparison between envelope-103 detected and RF-based echocardiographic speckle tracking 104 may be found in [34] and [38]. 105

As noted in [34], any of the above-mentioned approaches 106 faces an intrinsic limitation: the reduced lateral resolution (i.e. 107 in the direction perpendicular to the beam propagation axis) 108 and the absence of direct-phase information in the lateral 109 direction results in a low accuracy in the computation of lateral 110 displacements. Several algorithms have been described to 111 tackle this issue, based on sophisticated interpolation schemes 112 or re-correlation [4], [36], [39]. 113

Another way around consists in modifying the image for-114 mation in order to introduce phase information in the lateral 115 direction, *i.e.* by using a particular beam-forming step designed 116 to produce transverse oscillations. This approach has been ini-117 tiated by Jensen in the field of blood flow quantification [40], 118 [41] and in ultrasound elastography [42], [43]. Preliminary 119 results produced by our group recently extended this technique 120 to echocardiography [44]-[47]. 121

In this context, we describe in this paper a new setup for cardiac motion estimation, based on the following elements:

 a specific beamforming scheme for producing transverse oscillations (TO) in cardiac imaging, *i.e.* adapted to a sectorial acquisition geometry. As explained above, such approach allows introducing phase information in the lateral direction and thus improving accuracy of the 2D motion estimation in this direction.

a phase-based motion estimation algorithm specifically 130 dedicated to the obtained TO images. This estimator 131 locally constrains the motion to correspond to an affine 132 transform and exploits the available two-dimensional 133 phase of the TO images. Compared to previously pub-134 lished phase-based motion estimation methods (e.g. [45] 135 or [46]), the approach presented herein combines the 136 phases of two single-orthant analytical signals with an 137 affine transformation instead of simple translations. 138

¹³⁹ The accuracy of the proposed framework is evaluated *in*

silico from five ultra-realistic simulated sequences [48] mim-140 icking respectively three parasternal short-axis and two apical 141 four chamber acquisitions. The new estimator is contrasted 142 against other two phase-based estimators in [49] and [30] and 143 conventional block-matching applied to standard images (i.e. 144 without TO) [50]. Note that, although historically among the 145 earlier techniques proposed for motion estimation in medical 146 ultrasound [50], block-matching nonetheless still remains the 147 methodology of choice [35], [51]. 148

For each algorithm we evaluated the accuracy in recovering the simulated displacement field and in computing the cardiac strain. All algorithms exploiting transverse oscillations were found to estimate more accurately the lateral component of the displacement than standard block matching and this led to an overall better precision in the computation of the total displacement field and of the cardiac strain. Among the phase-based techniques considered the proposed algorithms was found to be in average the more accurate and reliable. An implementation of the new beamforming technique on an ultrasound research platform [52] is also presented along with a preliminary *in vivo* evaluation of the proposed motion estimation framework for the computation of cardiac strain on one healthy subject. Computed strain curves were in line with what reported in literature for an healthy heart.

The paper is structured as follows. Section II describes the generation of the transverse oscillations for sectorial cardiac acquisition. Section III presents the motion estimation algorithm and Section IV provides the details of the numerical experiments used to evaluate the proposed framework. Section V presents the obtained results, which are discussed in Section VI.

II. TRANSVERSE OSCILLATION ULTRASOUND IMAGES

Transverse oscillation (TO) ultrasound images exhibit in the 172 lateral direction the same kind of oscillations normally present 173 in the axial direction [41]. Lateral oscillations are obtained 174 by implementing a point spread function (PSF) presenting, in 175 addition to the common axial modulation, a modulation in 176 the lateral direction. TO modality was originally developed 177 for linear probes where, denoting by x and z lateral and axial 178 coordinates respectively, the system PSF can be written as [53] 179 $h(x,z) = h_x(x)h_z(z)$ with: 180

$$h_{\xi}(\xi) = e^{-\pi \frac{\xi^2}{\sigma_{\xi}^2}} \cos\left(2\pi \frac{\xi}{\lambda_{\xi}}\right) \tag{1}$$

where $\xi = \{x, z\}, \lambda_x (\lambda_z)$ is the lateral (axial) oscillation wavelength and $\sigma_x (\sigma_z)$ is the lateral (axial) full width half maximum (FWHM) of the Gaussian envelope [42].

The axial profile of the PSF is related to the excitation pulse 184 and the impulse response of the transducer elements used. As 185 a consequence, the axial modulation is naturally present in the 186 axial PSF profile and the weighting window can be adapted 187 using specific excitation pulses. The degrees of freedom that 188 enable one to control the transverse profile of the PSF are 189 instead the delay between the elements used in transmit and in 190 receive and the weighting coefficients applied to each element 191 in transmit and in receive. 192

149

150

151

152

153

154

155

156

157

158

159

160

161

162

163

164

165

166

167

168

169

170

171

181

182

To design these parameters the Fraunhofer approximation 193 is commonly used [54]. Fraunhofer approximation applies to 194 focused acoustic beams and states that, at the focal point, the 195 lateral beam profile $h_x(x)$ and the radiating aperture w(x) are 196 related by the Fourier transform [54]: 197

$$h_x(x_0) \propto \int_{\text{Aperture}} w(x) e^{-j\frac{2\pi}{\lambda_z z} x_0 x} dx$$
 (2)

From (2) it is straightforward to show that a bi-modal apodiza-198 tion function of the kind [42]: 199

$$w(x) \propto e^{-\pi \left(\frac{x-x_0}{\sigma_0}\right)^2} - e^{-\pi \left(\frac{x+x_0}{\sigma_0}\right)^2}$$
(3)

realizes the desired lateral profile, with $x_0 = z\lambda_z/\lambda_x$ and 200 $\sigma_0 = \sqrt{2}\lambda_z z/\sigma_x$. As these parameters depend on the axial 201 coordinate, the apodization function has to be dynamically 202 adjusted in order to obtain a depth-invariant PSF [53]. 203

A. Transverse oscillations in echocardiography 204

To date most consolidated applications of TO are blood flow 205 imaging [40], [41], [55] and elastography [42], [43]. In both 206 cases the presence of lateral oscillations has been shown to 207 favor a more accurate estimation of lateral displacements as 208 compared to traditional beamforming techniques. 209

Recently, the concept of TO has been extended to cardiac 210 ultrasound, where the accurate quantification of lateral heart 211 deformations still remains a challenge [56]. Our group had 212 a pioneering role in studying the feasibility of TO imaging 213 in echocardiography [44]–[47]. In particular, a beamforming 214 technique was presented in [44] for the generation of TO on 215 sectorial probes of common use in cardiac applications. The 216 beamformer design relies on the principle of back-propagation 217 and allows to obtain on pre-scan converted data (i.e. in polar 218 coordinates) the same kind of lateral oscillations otherwise 219 possible on linear probes. More specifically, a PSF completely 220 analogous to the one in (1) can be obtained in the polar space 221 (ρ, θ) . This is done according to the coordinate transformation 222 $\approx \rho$ and $x \approx \rho \theta$ and, consequently, the parameters z223 transformation $\lambda_x \approx \lambda_{\theta} \rho$, $\sigma_x \approx \sigma_{\theta} \rho$, $\lambda_z \approx \lambda_{\rho}$ and $\sigma_z \approx \sigma_{\rho}$. 224 The required apodization function has the same form as (3), 225 where the peaks position and width are given by $x_{\theta_0} = \lambda_{\rho}/\lambda_{\theta}$ 226 and $\sigma_{\theta_0} = \sqrt{2}\lambda_{\rho}/\sigma_{\theta}$. Interestingly, these quantities are no 227 longer depth-dependent as in linear geometries. As a result a 228 space invariant PSF can be obtained on sector scan without 229 dynamically modifying the apodization function. For more 230 detail we address the reader to [44]. 231

III. MOTION ESTIMATION ALGORITHM FOR 232 ECHOCARDIOGRAPHIC IMAGES WITH TRANSVERSE 233 OSCILLATIONS 234

Consider two rectangular blocks of pixels extracted from 235 two subsequent RF frames $s(x, z, t_0)$ and $s(x, z, t_1)$ (for 236 simplicity $t_1 = t_0 + 1$ of a cardiac ultrasound sequence. 23 The motion estimation problem consists in computing the 238 displacement field $\mathbf{d}(x,z) = [d_1(x,z), d_2(x,z)]^T$ mapping 239 the second block onto the first, being d_1 and d_2 the lateral 240 and axial components of the displacement respectively. This is 241

normally done by adopting the so called *brightness constancy* 242 assumption $s(x, z, t_0) = s(x - d_1(x, z), z - d_2(x, z), t_1).$

Nevertheless, it has been shown that brightness conservation 244 can be a misleading assumption as far as cardiac ultrasound 245 images are concerned [18]. The reason is that the amplitude 246 of the backscattered echo depends on the angle formed by 247 the acoustic beam and the myocardial fibers, which obviously varies in time due to the heart motion. As a consequence, 249 the same portion of tissue will return different echoes after it 250 position has changed. For this reason, we replace the classical 251 brightness constancy assumption with a more robust phase constancy assumption. Image phase is indeed ideally suited for 253 ultrasound images since it is independent on the local intensity 254 while intrinsically related to the local image structure.

In particular, 2D single-orthant analytic signals are used to 256 compute the image phase [57]. Based on multidimensional 257 Hilbert transforms, they represent one of the first attempts to 258 generalize the classical 1D analytic signal to n dimensions 259 (as for example for 2D images). The suitability of single-260 orthant analytic signals for modeling and processing TO 261 ultrasound images has been shown in different contexts from 262 the considered one in [49], [58]. 263

Based on the TO theory presented in the previous section, a signal model consisting of a 2D spatial modulation at spatial frequencies $1/\lambda_x$ and respectively $1/\lambda_z$ can be assumed [49]:

$$s(x, z, t_0) = w_s(x, z, t_0) \cos(2\pi x/\lambda_x) \cos(2\pi z/\lambda_z)$$
(4)

where w_s is a low-pass 2D window having its highest 264 frequency lower than the frequency of the 2D cosinus (a reasonable hypothesis in TO ultrasound imaging). 266

The four single-orthant analytic signals are then calculated by canceling three of the four quadrants in the 2D spectrum. However, given the symmetry of the 2D Fourier transform of real images, these analytic signals contain, two by two, redundant information [49]. For this reason, we only conserve two of the four available single-orthant analytic signals. Following [49], they can be expressed in the frequency domain as:

$$S_{so1}(f_1, f_2, t) = S(f_1, f_2, t)(1 + sign(f_1))(1 + sign(f_2))$$

$$S_{so2}(f_1, f_2, t) = S(f_1, f_2, t)(1 - sign(f_1))(1 + sign(f_2)).$$
(5)

where capitals letters denote the 2D Fourier transform, f_1 267 and f_2 denote the lateral and axial frequency respectively and 268 sign(x) = x/|x|. By denoting $\Phi_{so1}(x, z, t)$ and $\Phi_{so2}(x, z, t)$ 269 the spatial phases associated to the two considered analytic 270 signals, the phase constancy assumption reads as: 271

$$\begin{pmatrix} \Phi_{so1}(x, z, t_1) \\ \Phi_{so2}(x, z, t_1) \end{pmatrix} = \begin{pmatrix} \Phi_{so1}(x + d_1(x, z), z + d_2(x, z), t_0) \\ \Phi_{so2}(x + d_1(x, z), z + d_2(x, z), t_0) \end{pmatrix}.$$
(6)

Assuming small displacements, as commonly done in differ-272 ential techniques, the right hand side of (6) can be replaced by 273 its first order Taylor development and this leads to the linear 274 system of equations [49]: 275

$$\begin{pmatrix} \Phi_{so1}^{(t)}(x,z) \\ \Phi_{so2}^{(t)}(x,z) \end{pmatrix} = \mathbf{J} \cdot \begin{pmatrix} d_1(x,z) \\ d_2(x,z) \end{pmatrix}.$$
(7)

243

248

252

255

where $\Phi_{so2}^{(t)}(x,z)$ and $\Phi_{so2}^{(t)}(x,z)$ are the temporal deriva-276 tives of Φ_{so1} and Φ_{so2} respectively while 277

$$\mathbf{J}(x,z,t) = 2\pi \begin{pmatrix} 1/\lambda_x & 1/\lambda_z \\ -1/\lambda_x & 1/\lambda_z \end{pmatrix}$$
(8)

is the Jacobian matrix of the vector $[\Phi_{so1}, \Phi_{so2}]^T$. 278

While the motion estimation problem (7) could be in 279 principle solved pixel-wise, the corresponding solution would 280 be highly sensitive to noise, which is not acceptable in a 281 low SNR context as medical ultrasound. The common way 282 around this is to solve the problem in the least squares sense 283 by assuming that all the pixels in a block translate of the 284 same quantity, *i.e.* $\mathbf{d}(x, z) = \mathbf{d}_0$. Nevertheless, several studies 285 pointed out that the simple translation model is too restrictive 286 in the context of cardiac motion estimation [17], [18]. In this 287 scenario a much better solution is instead represented by the 288 affine model [17], [18]. Considering for simplicity a block 289 centered at $(x_0, z_0) = (0, 0)$, the affine model is: 290

$$\mathbf{d}(x,z) = \mathbf{A}(x,z)\mathbf{u}, \quad \mathbf{A} = \begin{bmatrix} 1 & 0 & x & z & 0 & 0 \\ 0 & 1 & 0 & 0 & x & z \end{bmatrix}, \quad (9)$$

where $\mathbf{u} = [d_{10}, d_{20}, d_{1x}, d_{1z}, d_{2x}, d_{2z}]^T$ is the new unknown 29 vector: d_{10} and d_{20} correspond to the translation of the window 292 center and $d_{ik} = \partial_k d_i$. 293

By plugging (9) into (7) and after suitable rearrangement of the system entries, it can be shown (see Appendix A) that the motion estimation problem can be independently solved for the two main directions x and z as:

$$\frac{\lambda_x}{4\pi} \begin{pmatrix} \Phi_{so1}^{(t)}(x_0, z_0) - \Phi_{so2}^{(t)}(x_0, z_0) \\ \dots \\ \Phi_{so1}^{(t)}(x_{N-1}, z_{N-1}) - \Phi_{so2}^{(t)}(x_{N-1}, z_{N-1}) \end{pmatrix} = \\
= \begin{pmatrix} 1 & x_0 & z_0 \\ \dots & \dots & \dots \\ 1 & x_{N-1} & z_{N-1} \end{pmatrix} \begin{pmatrix} d_{10} \\ d_{1x} \\ d_{1z} \end{pmatrix} \quad (10)$$

and

$$\frac{\lambda_z}{4\pi} \begin{pmatrix} \Phi_{so1}^{(t)}(x_0, z_0) + \Phi_{so2}^{(t)}(x_0, z_0) \\ \dots \\ \Phi_{so1}^{(t)}(x_{N-1}, z_{N-1}) + \Phi_{so2}^{(t)}(x_{N-1}, z_{N-1}) \end{pmatrix} = \\ = \begin{pmatrix} 1 & x_0 & z_0 \\ \dots & \dots & \dots \\ 1 & x_{N-1} & z_{N-1} \end{pmatrix} \begin{pmatrix} d_{20} \\ d_{2x} \\ d_{2z} \end{pmatrix} \quad (11)$$

where (x_k, z_k) $(k = 0, 1, \dots, N - 1)$ denotes the coordinate 294 of the k-th pixel of the considered block. 295

The two over-determined systems (10) and (11) are then 296 solved by classical least-squares fitting. We also remember 297 that given two complex numbers the sum of their phases is 298 equal to the phase of their product while the difference of their 299 phases is equal to the phase of the product of the first with 300 the conjugate of the second. These relations are better used 301 in the motion estimation framework to compute phase sums 302 and differences, since they allow avoiding tedious unwrapping 303 issues. 304

We also note that, since the phase of (4) does not change 305 for horizontal (vertical) shifts equal to the wavelength λ_x 306 (λ_z) , then the largest displacements that can be estimated 307

and (b) compare the same frame of the SAx3 sequence in the two acquisition modalities while (c) illustrates the M-mode computed over one cardiac cycle on the scan-line represented by the red segment. End systole (ES) has been assigned in correspondence of the highest muscle contraction and is denoted by a dark green line. For the SAx3 sequence ES corresponds roughly to frame 18

Fig. 1. Comparison of standard beamforming and transverse oscillations: (a)

unambiguously in the lateral and axial direction respectively are limited to $\lambda_x/2$ and $\lambda_z/2$.

Finally, while the motion estimation algorithm has been presented for Cartesian coordinates (x, z), as said in Section II, exactly the same considerations apply to pre-scan converted sectorial data simply by replacing (x, z) by the polar couple 313 $(\rho, \theta).$

IV. MATERIALS AND METHODS

A. Evaluation data set

A quantitative performance evaluation of the proposed framework was made in silico. A preliminary feasibility study in vivo will be shown in the results section.

For the in silico evaluation we made use of ultra-realistic 320 synthetic echocardiographic image sequences generated ac-321 cording to an original framework we recently developed [48]. 322 Briefly, cardiac motion is mimicked by displacing a set of 323 point scatterers over time. Both scatter amplitude and motion 324 are learned from a real echocardiographic acquisition adopted 325 as a template. From the time-variant scatter map FieldII is 326 employed to simulate the image formation process [59], [60]. 327 Since the synthetic cardiac motion is known, this can be used 328 to benchmark motion estimation algorithms. 329

The resulting synthetic sequences are extremely realistic 330 both in their motion and aspect, to the point it is hard to 331 distinguish them from real clinical recordings. In particular, 332 all the typical image artifacts such as reverberations, clutter 333 noise, signal dropout, local intensity and/or contrast variations 334 over time due to changing cardiac fiber orientation, which 335 have a major impact on the performance of algorithms for mo-336 tion/deformation estimation, are naturally present as they are 337 inherited from the template sequence [48]. As a consequence, 338 although obtained on synthetic data, the reported evaluation is 339 well representative of what could be expected on real clinical 340 data. 341

STANDARD TO <u>E</u> E 40 60 40 60 FRAME θ [°] θ [°] (a) (b) (c)

308

309

310

311

312

314

315

316

317

318

319

M-MODE



Fig. 2. Amplitude spectrum of the RF images corresponding to the two SAx frames in Fig. 1(a) and (b).

Five synthetic sequences were generated according to [48], 342 namely three parasternal short-axis (SAx) and two apical 343 four chamber (A4C), which correspond to two of the most 344 commonly employed views in the clinical practice [61]. In 345 the following the three SAx sequences will be denoted as 346 SAx1, SAx2 and SAx3 while the two A4C sequences will be 347 denoted as A4C1 and A4C2. For each of the five sequences 348 the two acquisition modalities of interest were simulated, *i.e.* 349 traditional beamforming and TO. Note that the time variant 350 scatter map associated to a synthetic sequence, and hence the 351 benchmark motion field, remained unchanged when modifying 352 the beamforming strategy. 353

The two acquisition strategies were implemented by suitably 354 setting the receive apodization function of the synthetic probe 355 used by FieldII: a standard Hanning window was used in the 356 standard case while the bimodal function in (3) was used in 357 the case with TO. In this latter case the parameters $\lambda_{\theta} = 6^{\circ}$, 358 $\sigma_{\theta} = 4\lambda_{\theta}$ were used, while the remaining probe settings 359 were kept constant for both acquisition modalities, namely 360 center frequency $f_0 = 4$ MHz, sampling frequency $f_s = 40$ 361 MHz, speed of sound c = 1540 m/s and 64 elements. Sweep 362 angle were 65° for the SAx sequences and 75° for the A4C 363 sequences. Frame rate was 50 frames/sec for the three SAx 364 sequences and 45 frames/sec for the two A4C ones. 365

Examples of simulated images are given in Fig. 1. Since 366 the TO model (1) holds for pre-scan converted data, images 367 are reported in the polar domain (ρ, θ) . It is evident how 368 the presence of lateral oscillations reflects in a richer speckle 369 pattern as compared to traditional beamforming. Fig. 1 only 370 allows assessing the visual realism of the individual frames. 371 In order to appreciate the dynamical behavior the reader is 372 addressed to the videos posted at http://www.creatis.insa-lyon. 373 fr/us-tagging/News_November_2011. 374

The Fourier spectrum of two simulated frames (radio frequency images are considered for the frequency analysis) obtained with standard beamforming and TO is reported in Fig. 2(a) and (b) respectively. Note the effect of the lateral modulation at the angular frequency $1/\lambda_{\theta}$ in the case with TO.

381 B. Cardiac motion estimation

As reference technique for standard RF images we will consider block matching (BM). On RF images with TO the estimator described in Section III, referred to as *Affine Phase Based Estimator* (APBE), will be contrasted against other two phase-based solutions: a previous version of the same estimator [49], referred to as *Translation Phase Based Estimator* (TPBE), where a simple translation model is considered instead of the affine model in (9), and a technique based on the maximizing block-wise the correlation between phase images, referred to as *Phase correlation estimator* (PhCorr). The latter, based on an iterative Newton algorithm, estimates the displacement by searching for the phase root of the complex cross-correlation function [30]. For one block of pixels, PhCorr method was implemented to estimate the displacements of all the columns and rows, and the final estimates were the mean values for each direction.

Considering RF data prevents axial down-sampling and this implies dealing with axial displacements which, at conventional sampling rates, are easily of the order of few tens of pixels. As an example, at the considered sampling rate ($f_s = 40$ MHz) and speed of sound (c = 1540 m/s), a displacement of 1 mm in the direction of the beam propagation would correspond to a shift of ~ 52 pixels.

Displacements of such entity violate the small displacements assumption essential in differential techniques as the proposed one. We dealt with this issue by proceeding in two steps: an initialization phase to produce a coarse estimate of the displacement and a successive refinement, where the proposed phase based estimator was applied to estimate the residual motion.

The initialization was performed with block-matching with sums of absolute differences as similarity criterion. At this stage no interpolation (*i.e.* no sub-pixel precision) was used in order to speedup the procedure. Initialization was not performed on the RF directly but on the B-mode. Indeed B-mode images, being base-band, are better suited for the analysis of large deformations than RF [38].

In order to have a fair comparison among the four considered algorithms, the same initialization was kept both when the refinement was made on standard RF (the case of BM) and on TO RF (the case of the APBE and TBPE algorithms). In particular the initial estimate was obtained from B-mode images without transverse oscillations.

The initialization procedure is summarized in Fig. 3. The 425 block-matching initialization $\mathbf{D} = [D_1, D_2]$ determines the 426 two blocks s_1 and s_2 for the successive refinement $\Delta \mathbf{D}$. The 427 total displacement is then given by $D+\Delta D$. For what concerns 428 the algorithm in Section III, the fact that the two blocks are not 429 aligned only implies replacing in (10) and (11) $\Phi_{so2}(x_i, z_i)$ 430 with $\Phi_{so2}(x_i + D_1, z_i + D_2)$. The parameters $[L_1, L_2]$ and 431 $[G_1, G_2]$ defined in Fig. 3 were the only required. They were 432 optimized in order to have the smallest estimation error (see 433 next section for more details) on the synthetic SAx3 sequence. 434 The optimal found values (in pixels) corresponded to: $L_1 =$ 435 16, $L_2 = 64$, $G_1 = 3$ and $G_2 = 64$. The RF image size was 436 4562×113 pixels². 437

When block-matching was employed for the refinement 438 interpolation factors of 1 and 6 were used in the axial and 439 lateral directions respectively. Note that, due to the high 440 sampling frequency of RF data, an interpolation of 1 in 441

384

385

386

387

388

389

390

391

392

393

394

395

396

397

398

399

400

401

402

403

404

405

406

407

408

409

410

411

412

413

414

415

416

417

418

419

420

421

422

423



Fig. 3. Initialization procedure. (a) Frame at time t. (b) Frame at time t + 1. The centers of the considered blocks are represented as black dots. The block s_1 is centered at (x_0, z_0) . D_1 and D_2 are the lateral and axial initial estimates given by block-matching and defining the center of the block s_2 . G_1 and G_2 define the spacing between the nodes. L_1 and L_2 define the block size. The successive refinement will be on s_1 and s_2 .

the axial direction was sufficient to obtain a resolution of 442 $c/(2f_s) \approx 0.02$ mm. 443

Let's finally point out that, since the TO image model holds 444 only for pre-scan converted data, in the proposed framework 445 motion estimation must be performed in polar coordinates. The 446 computed displacement field has then to be scan converted to 447 have the values in Cartesian coordinates. 448

C. Accuracy assessment 449

The four algorithms were compared in terms of accuracy in 450 the recovered displacement field and in the computed cardiac 451 strain. 452

1) Displacement field: Let's denote as $d_k(x, z)$ 453 $[\bar{d}_{k,1}(x,z), \bar{d}_{k,2}(x,z)]$ the ground-truth displacement between 454 frame k and frame k + 1 at position (x, z) and as $\mathbf{d}_k(x, z) =$ 455 $[d_{k,1}(x,z), d_{k,2}(x,z)]$ the estimated one. 456

The results in polar coordinates were considered first. 457 Hereto the main goal is to show the improvement in the 458 estimation of lateral displacements made possible thanks to 459 the proposed framework. Lateral and axial errors were used 460 for this purpose: 461

$$err_{k,lat}(x,z) = |d_{k,1}(x,z) - d_{k,1}(x,z)|,$$

$$err_{k,ax}(x,z) = |\bar{d}_{k,2}(x,z) - d_{k,2}(x,z)|,$$
(12)

where $|\cdot|$ denotes the absolute value. Error study was limited to 462 the region of the left-ventricle muscle. The latter was manually 463 contoured from the first frame of each synthetic sequence. The 464 mask was then propagated to all the frames of the sequence 465 by using the benchmark motion field. The value of these error 466 metrics will be reported in pixels. 467

Cardiac ultrasound sequences are commonly visualized and 468 processed in Cartesian coordinates, i.e. after scan conversion of 469 470 the beamformed images. For this reason the remaining part of the evaluation considered scan converted images. In this case 471 the accuracy in the recovered displacement was measured with 472 the endpoint error [62]: 473

$$EE_k(x,z) = ||\bar{\mathbf{d}}_k^{SC}(x,z) - \mathbf{d}_k^{SC}(x,z)||_2.$$
(13)

where $\bar{\mathbf{d}}_{k}^{SC}(x,z)$ and $\mathbf{d}_{k}^{SC}(x,z)$ denote the reference and the estimated displacement after scan conversion and $|| \cdot ||_2$ is the ℓ_2 -norm. Errors in Cartesian coordinates were measured in millimeters.

2) Strain: Cardiac strain was measured similarly to [63]. 478 The endocardium was first manually contoured in the ED 479 frame (i.e. the first frame of the sequence). A ROI for strain estimation was then created by expanding the endocardial contour along its normal to represent the myocardium. This region was subsequently populated in the directions normal and tangential to the endocardial contour with 6 and 100 sample points respectively, and given a label corresponding to one of the heart segments. Segments were established following the guidelines given by the American Heart Associated (AHA) [64]. Namely, six equally spaced segments around the circumference were considered for SaX views while, for what concerns apical views, three equally spaced longitudinal levels 490 were defined from base to apex, either on the septal or lateral 491 side, thus leading again to six segments. 492

The test points were then displaced over the full cardiac cy-493 cle by using the reference displacement and the displacement 494 estimated by each algorithm. The strain along a direction n at 495 time k was then computed as [29], [63]: 496

$$\epsilon_{\mathbf{n}}(k) = \frac{D_{\mathbf{n}}(k)}{D_{\mathbf{n}}(0)} - 1 \tag{14}$$

where $D_{\mathbf{n}}(k)$ denotes the distance between two consecutive 497 test points. More precisely, normal and tangential directions 498 on SaX sequences were used to determine radial and cir-499 cumferential strain components (ϵ_{RR} and ϵ_{CC}) respectively, 500 while the tangential directions on apical sequences was used 501 to determine the longitudinal component ϵ_{LL} . 502

Note that each simulated sequence corresponds to one 503 full cycle from one end-diastole to the following. Given the 504 periodicity of the cardiac cycle it is therefore reasonable to 505 assume $\epsilon_{\mathbf{n}}(N_F) = 0$ being N_F the number of frames in 506 the sequence. As in [29], [63], this condition is imposed by 507 applying the following drift compensation to the computed 508 strain curves. The strain compensated strain ϵ^{dc} is: 509

$$\epsilon_{\mathbf{n}}^{dc}(k) = \epsilon_{\mathbf{n}}(k) - \frac{k-1}{N_F - 1} \epsilon_{\mathbf{n}}(N_F).$$
(15)

Segmental strain was obtained by averaging the strain values computed point-wise on the test points on each segment.

3) Statistical analysis: For what concerns the accuracy in 512 retrieving the displacement field, the statistical significance of 513 the differences among the four algorithms was tested by means 514 of the Friedman rank test ($\alpha = 0.05$) in conjunction with the 515 post-hoc test proposed by Daniel [65], as suggested in [66]. 516 Strain accuracy was instead assessed by using the Pearson 517 correlation coefficient ρ together with the bias μ and standard 518 deviation σ returned by the Bland-Altman (BA) analysis. For 519 each correlation value the p-value was computed testing the 520

474

475

476

477

510

hypothesis of no correlation. The statistical significance of 521 each reported bias μ was measured with a t-test. Fisher's z-522 transform ($\alpha = 0.05$) was used to compare the strengths of 523 different correlations. T-test ($\alpha = 0.05$) was used to compare 524 the biases returned by the BA analysis. 525

Segmental strain values were considered and all segments 526 were included in the analysis. The three strain components 52 $\epsilon_{RR}, \epsilon_{CC}$ and ϵ_{LL} were considered independently. Among the 528 different phases in the cardiac cycle, the strain at end systole 529 has been shown to be particularly relevant for diagnosis [12]. 530 Hereto the accuracy in computing end-systolic strain values 531 was measured separately and will be presented in the results 532 section. Nevertheless, considering a single time instant reduced 533 the sample size to a point that statistical significance was 534 never observed. In order to have more statistically significant 535 results, and together to have a more exhaustive look at the 536 strain behaviour over time, the analysis was repeated by 537 including multiple time instants obtained by sampling one 538 frame out of eight. Temporal sub-sampling was adopted to 539 avoid correlation. 540

Let's finally note that no post processing operation such as 541 low-pass or median filtering was performed on the estimated 542 displacement fields. This was done to have a direct understand-543 ing of the relation between the data type (standard beamform-544 ing and TO) and the accuracy possible in the computation of 545 displacements. 546

547

549

V. RESULTS

A. In silico results 548

1) Accuracy in the displacement field:

Table I and Table II report respectively the lateral and axial 550 displacement errors measured in pixels on the five simulated 551 sequences before scan conversion while Table III reports the 552 errors measured in millimeters after scan conversion. Mean 553 values and standard deviations are computed for each sequence 554 by including all pixel estimates and all frames. An analysis of 555 the errors behaviour over time is provided by Fig. 4. 556

From Table I all algorithms exploiting TO images almost 55 consistently return in average more accurate lateral motion 558 estimates than regular block matching on standard RF images, 559 with the only exception of sequence A4C1 where the PhCorr 560 algorithm performs the worst. This result reveals that the 561 additional lateral information introduced by the TO frame-562 work can be effectively exploited to compute more accurate 563 estimates of cardiac motion. Moreover among the considered 564 algorithms the proposed APBE motion estimator regularly 565 produced the estimates with the smallest mean error and 566 the smallest variance, which reveals a superior accuracy and 567 reliability. As shown by Fig. 4(a)-(b), the higher accuracy 568 of the APBE method in the lateral direction was observed 569 at almost every time instant. 570

From what concerns errors in the axial direction, as shown 57 by Table II and Fig. 4(c)-(d), on one side both TPBE and Ph-572 Corr algorithms performed very similarly to BM. This is easily 573 explained by the fact that lateral oscillations do not modify 574 substantially the axial profile of the system PSF and hence no 575 576 improvement is expected for the motion computation in that



Fig. 4. Errors in the estimated displacement displayed over time for the SAx3 sequence. First, second and third row report the lateral error (in pixel), the axial error (in pixel) and the endpoint error after scan conversion (in millimeters) respectively. The mean error and its standard deviation are presented in adjacent subfigures. Each simulated cardiac cycle goes from end diastole to end diastole. On each error plot end systolic frame is illustrated by a vertical dashed line. For a more detailed understanding of the timing cf. the M-mode of the sequence in Fig. 1(c).

direction. On the other side the proposed APBE estimator was 577 observed to produce in average slightly higher errors than the 578 other algorithm considered. Note otherwise that, despite an 579 increase in the average error, the error dispersion obtained 580 by the proposed algorithm was still the smallest among the 581 considered techniques. As a remark note that the fact that 582 errors are much higher in the axial direction than in the lateral 583 one is explained by the fact that, as mentioned in Section 584 IV-B, axial displacement on RF data can be easily one order 585 of magnitude larger than later ones (few pixels vs. tenth of pixels).

The error after scan conversion is illustrated in Table III and Fig. 4(e)–(f). Clearly, is the value of displacement computed after scan conversion to represent the actual deformation of the cardiac muscle in the physical space and hence to have a meaning in the diagnostic process. In this case the proposed algorithm is the one returning the displacement estimates with 593 the smallest mean error and standard deviation. Again this 594 property is persistently observed at almost all time instants. 595

Concerning statistical consistency, all differences reported in

TABLE I Horizontal error (mean value \pm standard deviation). Values in pixels.

	SAx1	SAx2	SAx3	A4C1	A4C2
APBE	0.153 ± 0.139	0.131 ± 0.119	0.192 ± 0.186	0.307 ± 0.292	0.202 ± 0.174
TPBE	0.170 ± 0.167	0.147 ± 0.147	0.233 ± 0.232	0.365 ± 0.369	0.214 ± 0.192
BM	0.180 ± 0.216	0.206 ± 0.237	0.331 ± 0.365	0.440 ± 0.430	0.256 ± 0.290
PhCorr	0.148 ± 0.163	0.135 ± 0.151	0.208 ± 0.229	0.455 ± 0.447	0.204 ± 0.213

TARI F II Vertical error (mean value \pm standard deviation). Values in pixels.

	SAx1	SAx2	SAx3	A4C1	A4C2
APBE	1.0478 ± 2.2246	1.0114 ± 1.6374	2.0473 ± 3.4013	1.9055 ± 3.8254	0.5657 ± 1.7386
TPBE	0.8705 ± 2.3213	0.8669 ± 1.7177	1.8538 ± 3.5787	1.7467 ± 4.1863	0.4829 ± 1.9180
BM	0.9672 ± 2.3697	0.9150 ± 1.7659	1.8861 ± 3.6915	1.4724 ± 4.2151	0.5519 ± 1.9180
PhCorr	0.9027 ± 2.3535	0.8508 ± 1.7498	1.9344 ± 3.6270	5.9907 ± 7.3476	0.5795 ± 1.9456

TABLE III ENDPOINT ERROR (MEAN VALUE \pm STANDARD DEVIATION) AFTER SCAN CONVERSION. VALUES IN MILLIMETRES.

	SAx1	SAx2	SAx3	A4C1	A4C2
APBE	0.104 ± 0.112	0.095 ± 0.092	0.137 ± 0.146	0.119 ± 0.124	0.084 ± 0.091
TPBE	0.115 ± 0.133	0.105 ± 0.115	0.162 ± 0.178	0.133 ± 0.144	0.089 ± 0.103
BM	0.130 ± 0.177	0.151 ± 0.192	0.231 ± 0.279	0.174 ± 0.207	0.112 ± 0.167
PhCorr	0.105 ± 0.135	0.101 ± 0.124	0.150 ± 0.178	0.212 ± 0.220	0.085 ± 0.109



Fig. 5. Comparison between the errors computed by the proposed algorithm (in red) and the mean value of the reference displacement field (in green). Sub-figures (a), (b) and (c) correspond respectively to the errors in the lateral direction, axial direction and after scan conversion as reported in Fig. 4(a)(c)(e).

this section were find to be significant with p < 0.0001 as de-597 fined by the Friedman rank test. Note that spatial subsampling 598 of the displacement field was performed prior to the statistical 599 analysis in order to avoid correlation between samples. 600

Clearly, the value of the measured errors is correlated with 601 the velocity profile during the cardiac cycle: large errors are 602 expected in the instants of fastest motion as ejection and rapid 603 604 inflow while smaller errors are expected when the motion is



Fig. 6. Example of estimated motion fields on one diastolic (a) and one systolic (b) frame of the short axis sequence. The color encodes the radial velocity component according to the colormap in (c). The white cross denotes the LV center here located manually. Note how the estimated motion fields reflects the physiological expansion and contraction of the cardiac muscle in these two phases of the cardiac cycle.

slow as at end systole and end diastole. This explains the bi-605 modal behaviour of the error curves in Fig. 4. To give better 606 insights on this dependency, the error curves obtained with the proposed algorithm are put in relation with the average true displacement of the cardiac muscle in Fig. 5.

The spatial behaviour of the estimation error for the four approaches considered is reported in Fig. 7. The error image 611 is relative to the 27th frame of the simulated SAx3 sequence 612 and illustrates the performance of each algorithm in a worst 613 case scenario. Indeed at that instant, belonging to the rapid 614 ventricular filling phase (cf. Fig. 1(c)), the highest average 615 velocity over the entire cardiac cycle was measured. The error 616 maps confirm that all estimators based on TO RF images 617 outperform block matching in estimating lateral displacements 618



Fig. 7. Error images on frame 27 (maximum error) of the SAx3 sequence for the four considered algorithms: lateral error and axial error in the top and bottom row respectively. All errors reported in pixels. The sub-captions report the mean error \pm its standard deviation.

while the precision in the axial direction is very similar 619 among the four solutions. Among the four estimators, the 620 APBE algorithm produces the errors with the smallest average 62 value and the more uniform spatial distribution. Note that 622 the maximum error localized on the endocardial contour is 623 due to the motion of the mitral valve which interferes on the 624 displacement computation within the muscle. 625

For what concerns the BM algorithm, sub-pixel accuracy 626 was obtained in the lateral direction by interpolating of a 627 factor 6 while no interpolation was employed in the axial 628 direction. We verified that no relevant improvement in the 629 motion estimation accuracy was obtained by increasing those 630 values. 631

2) Strain analysis: Table IV compares the four algorithms 632 in terms of their accuracy in the computation of cardiac 633 strain. Multiple frames are considered in order to have a 634 statistically relevant comparison. All algorithms exploiting TO 635 were observed to return more accurate strain estimates for 636 all the three directions. The proposed APBE algorithm was 637 the one producing in average more consistent estimates: it 638 produced the highest correlation for the two strain components 639 ϵ_{RR} and ϵ_{LL} , the smallest bias for the two components ϵ_{RR} 640 and ϵ_{CC} and the smallest standard deviation for ϵ_{LL} . In the 641 remaining cases the TPBE algorithm was the most accurate, 642 however note that in those cases the differences with APBE 643 were not statistically significant, except for the bias of ϵ_{LL} . 644 Moreover note that measured biases for the APBE algorithm 645

were not statistically significant for ϵ_{RR} (p = 0.71) and ϵ_{CC} 646 (p = 0.72). The Bland-Altman plot of the four algorithms 647 considered for the radial, circumferential and longitudinal 648 strain components respectively are illustrated in Fig. 8, 9 and 649 10 respectively.

Examples of computed strain curves are provided in Fig. V-A1. All the three algorithms exploiting TO produce strain curves closer to the benchmark than BM. We measured the normalized distance between the estimated strain curve and 654 the benchmark as: 655

$$D_{algo} = \sqrt{\sum_{k} \frac{\left(\epsilon(k) - \bar{\epsilon}(k)\right)^2}{\bar{\epsilon}(k)^2}}$$
(16)

where $\bar{\epsilon}(k)$ is the benchmark global strain at time k while $\epsilon(k)$ 656 is the computed one. For all strain components, the APBE 657 algorithm returned the estimate with the smallest normalized 658 distance: for ϵ_{RR} it was $D_{APBE} = 0.19$, $D_{TPBE} = 0.23$, 659 $D_{\rm PhCorr}$ = 0.36 and $D_{\rm BM}$ = 0.85; for ϵ_{CC} it was $D_{\rm APBE}$ 660 = 0.10, D_{TPBE} = 0.16, D_{PhCorr} = 0.27 and D_{BM} = 0.67; for 661 ϵ_{LL} it was $D_{APBE} = 0.05$, $D_{TPBE} = 0.07$, $D_{PhCorr} = 0.91$ 662 and $D_{\rm BM} = 0.25$. 663

End-systolic strain values have been shown to be relevant 664 for the assessment of cardiac function. Given the size of the 665 data set, the number of end-systolic strain values were not 666 sufficient to have statistical significance. The results of the 667 strain analysis restricted to end-systole are reported for sake of 668

TABLE IV

Comparison of the four algorithms for estimation or radial strain ϵ_{RR} , circumferential strain ϵ_{CC} and longitudinal strain ϵ_{LL} in terms of correlation coefficient, Bland-Altman bias μ and Bland-Altman limits of agreement σ . The *p*-value between brackets reports the statistical significance of the reported value. The symbol * denotes values statistically different (p < 0.05) from the one of APBE. Statistical significance of the differences was measured with the Fisher's z-value for ρ and with a T-test for μ .

	ϵ_{RR}			ϵ_{CC}			ϵ_{LL}		
	ρ (<i>p</i> -value)	μ (<i>p</i> -value)	σ	ρ (<i>p</i> -value)	μ (<i>p</i> -value)	σ	ρ (<i>p</i> -value)	μ (<i>p</i> -value)	σ
APBE TPBE BM PhCorr	$\begin{array}{c} 0.96 \ (< 0.001) \\ 0.94 \ (< 0.001) \\ 0.64^{\star} \ (< 0.001) \\ 0.87^{\star} \ (< 0.001) \end{array}$	0.24 (0.71) 1.04 (0.13) 12.41* (<0.001) -0.69 (0.66)	6.60 6.71 25.37 10.45	0.82 (0.000) 0.85 (0.000) 0.50* (0.008) 0.81 (0.002)	$\begin{array}{c} 1.43 \ (0.72) \\ 2.06 \ (< \ 0.001) \\ 5.47^{\star} \ (< \ 0.001) \\ 2.42^{\star} \ (< \ 0.001) \end{array}$	5.00 4.10 9.47 5.11	0.94 (0.000) 0.93 (0.000) 0.088* (0.972) 0.86* (0.002)	$\begin{array}{c} -0.03 \ (< 0.001) \\ 0.02^{\star} \ (< 0.001) \\ 3.29^{\star} \ (0.002) \\ 0.86^{\star} \ (0.003) \end{array}$	1.29 1.44 6.63 1.83



Fig. 8. Bland Altman plot for the radial strain component ϵ_{RR} . The horizontal line in each plot represents the bias μ while the two dashed lines represent the limits of agreement $[\mu - 1.96\sigma; \mu + 1.96\sigma]$.



Fig. 9. Bland Altman plot for the circumferential strain component ϵ_{CC} . The horizontal line in each plot represents the bias μ while the two dashed lines represent the limits of agreement $[\mu - 1.96\sigma; \mu + 1.96\sigma]$.

completeness in Table V. What qualitatively shown is that also
in this case techniques based on TO in average outperform BM
(cf. biases and standard deviations). Comparing the different
phase-based techniques becomes instead more complicated
given the very limited number of samples.

3) Computational complexity: A final issue concerns the 674 computational complexity. Fast processing is indeed particu-675 larly desirable as far as medical ultrasound is concerned, since 676 the real-time capability is one of the main advantages of this 677 technique. All the considered algorithms were implemented 678 in MATLAB (R2011b, The Math-Works, Natick, MA) and 679 executed on a desktop PC with a 3.47 GHz Intel Xeon X5690 680 processor, 12 Gb of RAM and running Windows 7. The RF 681 image size was of 4562×112 pixels² for the SAx sequence 682 and 6724×189 pixels² for the A4C. 683

The most onerous step was the block-matching initialization, which on the SAx sequence took roughly 60 s/frame. This is clearly a limitation of the current implementation. Nevertheless one should consider that real-time implementa-687 tions of speckle-tracking exist and can be directly employed 688 to speed up the initialization procedure [67]. Concerning the 689 refinement instead, this took roughly 2.5 s/frame for the phase-690 based estimators and 22 s/frame for BM. Again, the reported 691 times are certainly implementation dependent. In particular 692 more effective implementation can be adopted to decrease the 693 cost associated to BM. Nonetheless it is important to note 694 the computational complexity of the phase based estimators 695 is considerably inferior to the one associated to BM. Indeed, 696 in the first case, the displacement is directly given by the 697 solution of the two 3×3 linear systems of equations given by 698 the least squares solution of (10) and (11), while BM requires 699 interpolation to obtain sub-pixel accuracy and the iterative 700 search of the best match position within each block. For this 701 reason a considerable speedup over BM can still be expected 702 even in more optimized implementations. 703



(c)

Fig. 10. Bland Altman plot for the longitudinal strain component ϵ_{LL} . The horizontal line in each plot represents the bias μ while the two dashed lines represent the limits of agreement $[\mu - 1.96\sigma; \mu + 1.96\sigma]$

(b)

TABLE V ${\tt Comparison of the four algorithms for strain estimation. Correlation coefficient, Bland-Altman bias \ \mu \ {\tt and Bland-Altman bias \ \mu \ and Bland-Altman \ bias \ \mu \ and \ bland-Altman \ bland-Al$ LIMITS OF AGREEMENT σ . STRAIN VALUES AT END SYSTOLE ARE CONSIDERED ONLY.

	ϵ_{RR}			ϵ_{CC}			ϵ_{LL}		
	ρ (<i>p</i> -value)	BA μ	BA σ	ρ	BA μ	BA σ	ρ (<i>p</i> -value)	BA μ	BA σ
APBE TPBE BM PhCorr	0.94 (0.001) 0.91 (0.001) 0.92 (0.003) 0.90 (0.003)	-0.22 -0.30 4.50 -4.66	9.79 8.64 25.11 14.81	0.95 (0.00) 0.97 (0.00) 0.85 (0.00) 0.91 (0.00)	1.58 2.57 4.94 3.01	7.37 5.53 9.88 7.45	0.86 (0.14) 0.87 (0.13) 0.76 (0.24) 0.90 (0.09)	-0.13 0.12 5.17 0.64	2.55 2.89 9.99 2.29

B. In vivo results 704

(%)

ror

(a)

The goal to this section is to show that the state of advance 705 of the proposed framework is beyond simple simulation and an 706 in vivo evaluation on real clinical recording is already possible. 707

The proposed image formation technique with transverse 708 oscillations was implemented on a real scanner. In particular 709 the ultrasound research platform Ula-op [52] equipped with a 710 cardiac probe model PA230 from Esaote (Esaote Spa, Genoa, 711 Italy) was used. The acquisitions was performed by an experi-712 enced radiologist on one 25 years old male healthy volunteer. 713 In particular two views were acquired: one apical four chamber 714 and one parasternal short axis. In order to compare standard 715 B-mode images and the proposed TO beamforming, the RF 716 lines of the two imaging modalities were interleaved during the 717 acquisition: every second line corresponded to a conventional 718 B-mode sector scan and the other one used the beamforming 719 strategy of Section II to provide TO images. The frame rate 720 for both modalities was of 25 frames/s. The RF signals were 721 acquired at a sampling frequency of 50 MHz. The beam 722 density was of 1 beam/degree. Due to memory limitations, 723 a total of 49 frames (2 seconds) for each mode could be 724 acquired. This was sufficient to obtain one complete cardiac 725 cycle. 726

Fig. 12 shows one sample frame from each of the two views 727 when acquired with and without TO. Fig. 13 shows a selection 728 corresponding to the heart septum on the short axis view. From 729 the latter the difference in speckle pattern is evident. 730

Both sequences were processed with the APBE algorithm of 73 Section III. The strain curves for the radial and circumferential 732 strain components computed from the SAx sequence are 733 reported in Fig. 14. The measured strain values are consistent 734 with what reported in literature for an healthy heart [68]. A 735 thorough clinical evaluation including strain and strain rate 736

values on healthy and pathological subjects falls beyond the 737 scope of this paper and will be made object of future studies. 738

(d)

VI. DISCUSSIONS AND CONCLUSION

The paper introduced a novel setup for improving cardiac 740 motion estimation with ultrasound. Despite the important 741 progresses made in the field even best performing techniques 742 still register a low accuracy in estimating displacement/strain 743 values in the lateral direction (*i.e.* perpendicular to the beam 744 propagation). The proposed framework aimed at overcoming this limitation by combining two elements: an unconventional beamforming technique and a dedicated motion estimation algorithm.

The beamformer was designed so to add oscillations in the 749 lateral direction. As already known from blood flow imaging 750 and elastography such an acquisition scheme leads to an image 751 model intrinsically better suited for the estimation of lateral 752 displacements. We then presented an algorithm specifically 753 designed to exploit the availability of a phase information 754 in the two directions. This was done by decomposing the 755 ultrasound image into two 2D single-orthant analytic signals 756 and assuming time conservation of the two associated image 757 phases. 758

A quantitative evaluation of the proposed setup was per-759 formed in silico on five synthetic cardiac ultrasound sequences. 760 The comparison included block-matching on standard images 761 without transverse oscillations and other two phase-based 762 solutions exploiting the presence of oscillations in the lateral 763 direction. The obtained results revealed an higher accuracy in 764 the estimation of the cardiac motion when TO were employed. 765 In particular the proposed estimator were the most accurate 766 among the three phase-based algorithms. This better accuracy 767 reflected into a more robust estimation of the cardiac strain. 768

739



Fig. 11. Comparison among the four computed strain curves and the benchmark. Global strain values (i.e. averaged over the entire muscle) are considered at each time instant.

More specifically the proposed setup was the one allowing in 769 average for the highest correlation with the reference strain 770 values, the smallest bias and the smallest limits of agreement 77 as computed with the Bland Altman analysis. 772

While the estimation of lateral displacements were improved 773 by the employment of the affine model, leading to an overall 774 higher accuracy on the motion field after scan-conversion, we 775 acknowledge that the estimates in the axial direction were 776 in average slightly less precise than what obtained with the 777 other techniques considered. A possible reason is that a more 778 complex model (as the affine one) is more prone to over-fitting 779 than a simple one (as the translation one) in the presence of 780 noise. One solution would be locally choosing for the model 781 best adapted to the data (translation or affine in our case) as 782 proposed e.g. in [69]. This possibility will be considered in 783 future studies. 784

The synthetic evaluation is a first necessary step towards 785 a more thorough validation including phantom experiments 786 and real patients, which will be the topic of future studies. 787 In this perspective an implementation of the proposed beam-788 789 forming technique with transverse oscillations on the UlaOP



Sample images acquired with the UlaOp platform. The two Fig. 12. acquisition modalities on an apical view are compared in (a) and (b). The two acquisition modalities on an short axis view are compared in (c) and (d).



(b) with TO

Fig. 13. Zoom of the heart septum on the short axis view without(a) and with (b) transverse oscillations.

ultrasound research platform was presented in the paper along with the preliminary motion estimation results on one healthy volunteer. In particular in this preliminary evaluation it was shown that the extracted strain curves were consistent with what expected from the literature.

Concerning the computational complexity, the generation of transverse oscillations only implies modifying the receive apodization function of the system and hence it does not increase the computational demand. Instead for what concerns

798

790

791



Fig. 14. Strain curves computed with the proposed APBE algorithm on the short-axis acquisition.

the motion estimation algorithm its implementation in terms of 799 computational efficiency is still sub-optimal (MATLAB imple-800 mentation) and hence not competitive with the block matching 801 implementations present on commercial systems. Nevertheless 802 the proposed estimator is in principle less onerous than block 803 matching as no iterative research within a search window 804 is needed and no interpolation is needed to reach sub-pixel 805 accuracy. Moreover, being local, the proposed estimator is 806 intrinsically parallelizable and can hence take advantage of 80 parallel computation platforms as GPUs. 808

An issue that requires consideration in view of a clinical 809 evaluation is how transverse oscillations are perceived by the 810 final user, *i.e.* the physician. This evaluation must consider the 811 opinion of multiple experts and falls beyond the scope of this 812 paper. Nevertheless one should consider that several possibil-813 ities exist to exploit transverse oscillations for motion/strain 814 estimation while visualizing images close to the standard b-815 mode images currently of use in the clinical practice. One 816 possibility would be to acquire the two modalities in parallel 817 (possibly with a dedicated architecture for TO). The second 818 possibility would be extending envelope detection to the lateral 819 direction so to account for lateral oscillations. Interestingly this 820 2D envelope could be directly obtained as the amplitude of 821 the single-orthant analytic signal computed in (5), and hence 822 would not require supplemental calculations. 823

Future studies include an extension of the proposed setup to 824 3D echocardiography. Despite 2D still remains the modality 825 of choice in the clinical practice, 3D US has shown to be 826 potentially more accurate in the quantification of cardiac 827 mechanics and, therefore, a more reliable diagnostic tool. For 828 what concerns the proposed framework, the extension of the 829 estimator to 3D is straightforward, cf. [49]. For the beamform-830 ing of 3D TO images, several approaches are possible. First of 831 all a matrix array will be necessary because a two dimensional 832 apodization function must be designed. Pihl and Jensen and 833 Pihl et al. proposed in [70] and in [71] respectively a twofold 834 2D approach where two different 3D volumes are formed to 835 estimate 3D vector motion maps: one with TO oriented in the 836 lateral direction and one with the TO oriented in the elevation 837 direction. In [72], Salles et al. proposed to use instead a 838 separable 2D apodization function featuring 4 Gaussian peaks 839 to obtain directly volumes featuring both, lateral and elevation 840

oscillations.

VII. ACKNOWLEDGMENT

This work was supported by the French National Research 843 Agency (ANR) under the US-Tagging Grant and partially 844 supported by ANR-11-LABX-0040-CIMI within the program 845 ANR-11-IDEX-0002-02 of the University of Toulouse. It was 846 performed within the framework of the LABEX PRIMES 847 (ANR-11-LABX-0063) of Université de Lyon, within the 848 program "Investissements d'Avenir" (ANR-11-IDEX-0007) 849 operated by ANR. 850

Appendix

For one pixel of coordinates (x, z), using the analytical expression of the Jacobian matrix given in (8), the spatial phase time consistency in (7) may be further developped as a system of two equations. 855

$$\Phi_{so1}^{(t)}(x,z) = 2\pi/\lambda_x d_1(x,z) + 2\pi/\lambda_z d_2(x,z)$$

$$\Phi_{so2}^{(t)}(x,z) = -2\pi/\lambda_x d_1(x,z) + 2\pi/\lambda_z d_2(x,z)$$
(17)

Replacing $d_1(x, z)$ and $d_2(x, z)$ in (17) by the affine model in (9) and adding and substracting the two previous equations leads to: 858

$$\lambda_{x}/2\pi(\Phi_{so1}^{(t)}(x,z) - \Phi_{so2}^{(t)}(x,z)) = \begin{pmatrix} 1 & x & z \end{pmatrix} \begin{pmatrix} d_{10} \\ d_{1x} \\ d_{1z} \end{pmatrix}$$
$$\lambda_{z}/2\pi(\Phi_{so1}^{(t)}(x,z) + \Phi_{so2}^{(t)}(x,z)) = \begin{pmatrix} 1 & x & z \end{pmatrix} \begin{pmatrix} d_{20} \\ d_{2x} \\ d_{2z} \end{pmatrix} (18)$$

Finally, by applying (18) for a block of N pixels of coordinates (x_i, z_i) with i running from 0 to N-1, we obtain the two systems of equations given in (10) and (11).

REFERENCES

- [1] W. H. Organization, World health statistics 2012, 2012.
- [2] T. P. Abraham, V. L. Dimaano, and H.-Y. Liang, "Role of tissue doppler and strain echocardiography in current clinical practice," *Circulation*, vol. 116, no. 22, pp. 2597–2609, 2007.
- [3] C. Cottrell and J. N. Kirkpatrick, "Echocardiographic strain imaging and its use in the clinical setting," *Expert Review of Cardiovascular Therapy*, vol. 8, no. 1, pp. 93–102, 2010.
- [4] J. D'hooge, E. Konofagou, F. Jamal, A. Heimdal, L. Barrios, B. Bijnens, J. Thoen, F. Van de Werf, G. Sutherland, and P. Suetens, "Twodimensional ultrasonic strain rate measurement of the human heart in vivo," *Ultrasonics, Ferroelectrics and Frequency Control, IEEE Transactions on*, vol. 49, no. 2, pp. 281–286, 2002.
- [5] A. M. Shah and S. D. Solomon, "Myocardial deformation imaging: Current status and future directions," *Circulation*, vol. 125, no. 2, pp. e244–e248, 2012.
- [6] G. R. Sutherland, G. Di Salvo, P. Claus, J. D'Hooge, and B. Bijnens, "Strain and strain rate imaging: a new clinical approach to quantifying regional myocardial function," *Journal of the American Society of Echocardiography*, vol. 17, no. 7, pp. 788–802, 2004.
- [7] G. E. Mailloux, A. Bleau, M. Bertrand, and R. Petitclerc, "Computer analysis of heart motion from two-dimensional echocardiograms," *Biomedical Engineering, IEEE Transactions on*, vol. BME-34, no. 5, pp. 356–364, 1987.

13

841

842

851

863 864 865 866 867 868 869 870 871 872 873 874 875 876 877 878 879 880

881

882

883

884

885

- [8] M. De Craene, G. Piella, O. Camara, N. Duchateau, E. Silva, A. Doltra, 886 J. Dhooge, J. Brugada, M. Sitges, and A. F. Frangi, "Temporal dif-887 feomorphic free-form deformation: Application to motion and strain estimation from 3d echocardiography," *Medical Image Analysis*, vol. 16, 888 889 no. 2, pp. 427-450, 2012. 890
- B. Heyde, R. Jasaityte, D. Barbosa, V. Robesyn, S. Bouchez, P. Wouters, [9] 891 F. Maes, P. Claus, and J. D'hooge, "Elastic image registration versus 892 speckle tracking for 2-d myocardial motion estimation: A direct com-893 parison in vivo," Medical Imaging, IEEE Transactions on, vol. 32, no. 2, 894 pp. 449-459, 2013. 895
- [10] M. Alessandrini, H. Liebgott, D. Barbosa, and O. Bernard, Monogenic 896 Phase Based Optical Flow Computation for Myocardial Motion Analysis 897 898 in 3D Echocardiography, ser. Lecture Notes in Computer Science. Springer Berlin Heidelberg, 2013, vol. 7746, ch. 19, pp. 159-168. 899
- [11] K. McLeod, A. Prakosa, T. Mansi, M. Sermesant, and X. Pennec, 900 An Incompressible Log-Domain Demons Algorithm for Tracking Heart 901 Tissue, ser. Lecture Notes in Computer Science. Springer Berlin 902 Heidelberg, 2012, vol. 7085, ch. 6, pp. 55-67. 903
- M. De Craene, S. Marchesseau, B. Heyde, H. Gao, M. Alessandrini, [12] 904 O. Bernard, G. Piella, A. Porras, L. Tautz, A. Hennemuth, A. Prakosa, 905 H. Liebgott, O. Somphone, P. Allain, S. Makram Ebeid, H. Delingette, 906 M. Sermesant, J. D'hooge, and E. Saloux, "3d strain assessment in ultra-907 908 sound (straus): A synthetic comparison of five tracking methodologies,' Medical Imaging, IEEE Transactions on, vol. 32, no. 9, pp. 1632-1646, 909 910 2013.
- [13] B. Touil, A. Basarab, P. Delachartre, O. Bernard, and D. Friboulet, 911 "Analysis of motion tracking in echocardiographic image sequences: 912 913 influence of system geometry and point-spread function," Ultrasonics, 914 vol. 50, no. 3, pp. 373-386, 2010.
- [14] P. Baraldi, A. Sarti, C. Lamberti, A. Prandini, and F. Sgallari, "Evalua-915 tion of differential optical flow techniques on synthesized echo images,' 916 Biomedical Engineering, IEEE Transactions on, vol. 43, no. 3, pp. 259-917 918 272, 1996.
- [15] G. E. Mailloux, F. Langlois, P. Simard, and M. Bertrand, "Restoration of 919 the velocity field of the heart from two-dimensional echocardiograms,' 920 Medical Imaging, IEEE Transactions on, vol. 8, no. 2, pp. 143-153, 921 1989 922
- [16] V. Behar, D. Adam, P. Lysyansky, and Z. Friedman, "The combined 923 924 effect of nonlinear filtration and window size on the accuracy of 925 tissue displacement estimation using detected echo signals," Ultrasonics, vol. 41, no. 9, pp. 743-753, 2004. 926
- M. Suhling, M. Arigovindan, C. Jansen, P. Hunziker, and M. Unser, [17] 927 928 "Myocardial motion analysis from b-mode echocardiograms," Image 929 Processing, IEEE Transactions on, vol. 14, no. 4, pp. 525-536, 2005.
- M. Alessandrini, A. Basarab, H. Liebgott, and O. Bernard, "Myocardial 930 [18] 931 motion estimation from medical images using the monogenic signal,' Image Processing, IEEE Transactions on, vol. 22, no. 3, pp. 1084–1095, 932 2013 933
- [19] V. Behar, D. Adam, P. Lysyansky, and Z. Friedman, "Improving mo-934 tion estimation by accounting for local image distortion," Ultrasonics, 935 vol. 43, no. 1, pp. 57-65, 2004. 936
- Q. Duan, E. Angelini, S. Homma, and A. Laine, "Validation of optical-[20] 937 flow for quantification of myocardial deformations on simulated rt3d 938 ultrasound," in Biomedical Imaging: From Nano to Macro, 2007. ISBI 939 2007. 4th IEEE International Symposium on, 2007, pp. 944-947. 940
- C. Kontogeorgakis, M. Strintzis, N. Maglaveras, and I. Kokkinidis, "Tu-941 [21] mor detection in ultrasound b-mode images through motion estimation 942 using a texture detection algorithm," in Computers in Cardiology 1994, 943 1994, pp. 117-120. 944
- 945 [22] F. Yeung, S. F. Levinson, and K. J. Parker, "Multilevel and motion model-based ultrasonic speckle tracking algorithms," Ultrasound in 946 Medicine & Biology, vol. 24, no. 3, pp. 427-442, 1998. 947
- M. Strintzis and I. Kokkinidis, "Maximum likelihood motion estimation 948 [23] in ultrasound image sequences," Signal Processing Letters, IEEE, vol. 4, 949 no. 6, pp. 156-157, 1997. 950
- Y. Yue, J. Clark, and D. Khoury, "Speckle tracking in intracar-951 [24] diac echocardiography for the assessment of myocardial deformation," 952 Biomedical Engineering, IEEE Transactions on, vol. 56, no. 2, pp. 416-953 425, 2009. 954
- [25] B. Cohen and I. Dinstein, "New maximum likelihood motion estimation 955 schemes for noisy ultrasound images," Pattern Recognition, vol. 35, 956 no. 2, pp. 455-463, 2002. 95
- [26] M. Linguraru, N. Vasilyev, P. del Nido, and R. Howe, "Fast block flow 958 tracking of atrial septal defects in 4d echocardiography," Medical Image 959 Analysis, vol. 12, no. 4, pp. 397-412, 2008. 960
- M. Ledesma-Carbayo, J. Kybic, M. Desco, A. Santos, M. Suhling, 961 [27] P. Hunziker, and M. Unser, "Spatio-temporal nonrigid registration for 962

ultrasound cardiac motion estimation," Medical Imaging, IEEE Transactions on, vol. 24, no. 9, pp. 1113-1126, 2005.

- J. Woo, B.-W. Hong, C.-H. Hu, K. K. Shung, C. C. J. Kuo, and [28] P. Slomka, "Non-rigid ultrasound image registration based on intensity and local phase information," Journal of Signal Processing Systems, vol. 54, no. 1-3, pp. 33-43, 2009.
- [29] A. Elen, C. Hon Fai, D. Loeckx, G. Hang, P. Claus, P. Suetens, F. Maes, and J. D'Hooge, "Three-dimensional cardiac strain estimation using spatio-temporal elastic registration of ultrasound images: A feasibility study," Medical Imaging, IEEE Transactions on, vol. 27, no. 11, pp. 1580-1591, 2008.
- [30] A. Pesavento, C. Perrey, M. Krueger, and H. Ermert, "A time-efficient and accurate strain estimation concept for ultrasonic elastography using iterative phase zero estimation," IEEE Transactions on Ultrasonics, Ferroelectrics and Frequency Control, vol. 46, no. 5, pp. 1057-1067, 1999
- [31] M. Lubinski, S. Emelianov, and M. ODonnell, "Speckle tracking methods for ultrasonic elasticity imaging using short-time correlation," Ultrasonics, Ferroelectricity and Frequency Control, IEEE Transactions on, vol. 46, no. 1, pp. 82-96, 1999.
- [32] H. Chen, H. Shi, and T. Varghese, "Improvement of elastographic displacement estimation using a two-step cross-correlation method,' Ultrasound in Medicine & Biology, vol. 33, no. 1, pp. 48-56, 2007.
- J. D'hooge, B. Bijnens, J. Thoen, F. Van de Werf, G. Sutherland, and [33] P. Suetens, "Echocardiographic strain and strain-rate imaging: A new tool to study regional myocardial function," Medical Imaging, IEEE Transactions on, vol. 21, no. 9, pp. 1022-1030, 2002.
- R. G. P. Lopata, M. M. Nillesen, H. H. G. Hansen, I. H. Gerrits, J. M. [34] Thijssen, and C. L. de Korte, "Performance evaluation of methods for two-dimensional displacement and strain estimation using ultrasound radio frequency data," Ultrasound in Medicine & Biology, vol. 35, no. 5, pp. 796-812, 2009.
- [35] R. G. Lopata, M. M. Nillesen, J. M. Thijssen, L. Kapusta, and C. L. de Korte, "Three-dimensional cardiac strain imaging in healthy children using rf-data," Ultrasound in Medicine & Biology, vol. 37, no. 9, pp. 1399 - 1408, 2011.
- E. Konofagou and J. Ophir, "A new elastographic method for estimation [36] and imaging of lateral displacements, lateral strains, corrected axial strains and poissons ratios in tissues," Ultrasound in Medicine & Biology, vol. 24, no. 8, pp. 1183 - 1199, 1998.
- [37] J. Luo and E. Konofagou, "High-frame rate, full-view myocardial elastography with automated contour tracking in murine left ventricles in vivo," Ultrasonics, Ferroelectricity and Frequency Control, IEEE Transactions on, vol. 55, no. 1, pp. 240-248, 2008.
- [38] W. Yu, P. Yan, A. Sinusas, K. Thiele, and J. Duncan, "Towards pointwise motion tracking in echocardiographic image sequences - comparing the reliability of different features for speckle tracking," Medical Image Analysis, vol. 10, no. 4, pp. 495-508, 2006.
- M. A. Lubinski, S. Y. Emelianov, K. R. Raghavan, A. E. Yagle, A. R. [39] Skovoroda, and M. O'Donnell, "Lateral displacement estimation using tissue incompressibility," Ultrasonics, Ferroelectrics and Frequency Control, IEEE Transactions on, vol. 43, no. 2, pp. 247-256, 1996.
- [40] M. Aderson, "Multi-dimensional velocity estimation with ultrasound using spatial quadrature," Ultrasonics, Ferroelectrics and Frequency Control, IEEE Transactions on, vol. 45, no. 3, pp. 852-861, may 1998.
- J. Jensen and P. Munk, "A new method for estimation of veloc-[41] ity vectors," Ultrasonics, Ferroelectrics and Frequency Control, IEEE 1019 Transactions on, vol. 45, no. 3, pp. 837 -851, may 1998.
- [42] H. Liebgott, J. Wilhjehm, J. Jensen, D. Vray, and P. Delachartre, "Psf dedicated to estimation of displacement vectors for tissue elasticity imaging with ultrasound," Ultrasonics, Ferroelectrics and Frequency Control, IEEE Transactions on, vol. 54, no. 4, pp. 746-756, april 2007.
- [43] C. Sumi, "Displacement vector measurement using instantaneous ultrasound signal phase-multidimensional autocorrelation and doppler methods," Ultrasonics, Ferroelectrics and Frequency Control, IEEE Transactions on, vol. 55, no. 1, pp. 24-43, january 2008.
- X. Guo, D. Friboulet, and H. Liebgott, "Transverse oscillations beam-[44] former design for sector scan using back-propagation," in Biomedical Imaging (ISBI), 2012 9th IEEE International Symposium on, may 2012, pp. 1100 -1103.
- [45] H. Liebgott, A. Basarab, S. Marincas, O. Bernard, and D. Friboulet, "Tangential oscillations for motion estimation in echocardiography," in Ultrasonics Symposium, 2008. IUS 2008. IEEE, nov. 2008, pp. 1761 -1764
- H. Liebgott, A. Ben Salem, A. Basarab, H. Gao, P. Claus, J. D'hooge, [46] P. Delachartre, and D. Friboulet, "Tangential sound field oscillations for

963

964

965

966

1006

1007

1008

1009

1010

1011

1012

1013

1014

1015

1016

1017

1018

1020

1021

1022

1023

1024

1025

1026

1027

1028

1029

1037

2d motion estimation in echocardiography," in Ultrasonics Symposium (IUS), 2009 IEEE International, sept. 2009, pp. 498 -501.

[47] M. Alessandrini, H. Liebgott, D. Friboulet, and O. Bernard, "Monogenic 1041 phase based myocardial motion analysis from cardiac ultrasound with 1042 transverse oscillations," in Ultrasonics Symposium (IUS), 2012 IEEE 1043 International, 2012, in press. 1044

1039

1040

- 1045 [48] , "Simulation of realistic echocardiographic sequences for groundtruth validation of motion estimation," in Image Processing (ICIP), 2012 1046 19th IEEE International Conference on, 30 2012-Oct. 3, pp. 2329-2332. 1047
- A. Basarab, H. Liebgott, and P. Delachartre, "Analytic estimation of 1048 [49] subsample spatial shift using the phases of multidimensional analytic 1049 signals," Image Processing, IEEE Transactions on, vol. 18, no. 2, pp. 1050 1051 440-447. Feb.
- [50] L. Bohs and G. Trahey, "A novel method for angle independent 1052 ultrasonic imaging of blood flow and tissue motion," Biomedical En-1053 gineering, IEEE Transactions on, vol. 38, no. 3, pp. 280-286, March. 1054
- W.-N. Lee, Z. Oian, C. L. Tosti, T. R. Brown, D. N. Metaxas, and E. E. 1055 [51] 1056 Konofagou, "Preliminary validation of angle-independent myocardial Ultrasound in elastography using mr tagging in a clinical setting," 1057 Medicine & Biology, vol. 34, no. 12, pp. 1980 - 1997, 2008. 1058
- [52] P. Tortoli, L. Bassi, E. Boni, A. Dallai, F. Guidi, and S. Ricci, "Ula-1059 op: an advanced open platform for ultrasound research," Ultrasonics, 1060 1061 Ferroelectrics and Frequency Control, IEEE Transactions on, vol. 56, no. 10, pp. 2207-2216, october 2009. 1062
- [53] H. Liebgott, A. Basarab, P. Gueth, D. Friboulet, and P. Delachartre, 1063 "Transverse oscillations for tissue motion estimation," Ultrasonics, 1064 vol. 50, no. 6, pp. 548 - 555, 2010. 1065
- 1066 [54] J. W. Goodman, Introduction to Fourier Optics, 3rd ed. Roberts & 106 Company Publishers, Dec. 2004.
- [55] K. Hansen, J. Udesen, C. Thomsen, J. Jensen, and M. Nielsen, "In vivo 1068 validation of a blood vector velocity estimator with mr angiography,' 1069 Ultrasonics, Ferroelectrics and Frequency Control, IEEE Transactions 1070 on, vol. 56, no. 1, pp. 91 -100, january 2009. 1071
- J. D'hooge, A. Heimdal, F. Jamal, T. Kukulski, B. Bijnens, F. Rade-1072 [56] makers, L. Hatle, P. Suetens, and G. R. Sutherland, "Regional Strain 1073 and Strain Rate Measurements by Cardiac Ultrasound: Principles, Im-1074 plementation and Limitations," European Journal of Echocardiography, 1075 vol. 1, no. 3, pp. 154-170, Sep. 2000. 1076
- T. Bulow and G. Sommer, "Hypercomplex signals-a novel extension 1077 [57] 1078 of the analytic signal to the multidimensional case," Signal Processing, IEEE Transactions on, vol. 49, no. 11, pp. 2844-2852, 2001. 1079
- A. Basarab, P. Gueth, H. Liebgott, and P. Delachartre, "Phase-based [58] 1080 block matching applied to motion estimation with unconventional beam-1081 1082 forming strategies," Ultrasonics, Ferroelectrics and Frequency Control, IEEE Transactions on, vol. 56, no. 5, pp. 945-957, 2009. 1083
- [59] J. Jensen and N. Svendsen, "Calculation of pressure fields from arbitrar-1084 ily shaped, apodized, and excited ultrasound transducers," Ultrasonics, 1085 Ferroelectrics and Frequency Control, IEEE Transactions on, vol. 39, 1086 1087 no. 2, pp. 262-267, March 1992.
- [60] J. A. Jensen, "Field: A program for simulating ultrasound systems," 1088 1089 in 10th Nordic-Baltic Conference on Biomedical Imaging, vol. 34, Supplement 1. Medical & Biological Engineering & Computing, 1996, 1090 pp. 351-353. 1091
- T. Dietenbeck, M. Alessandrini, D. Barbosa, J. D'hooge, D. Fri-1092 [61] boulet, and O. Bernard, "Detection of the whole myocardium in 2d-1093 echocardiography for multiple orientations using a geometrically con-1094 strained level-set," Medical Image Analysis, vol. 16, no. 2, pp. 386 -1095 401 2012 1096
- [62] M. Otte and H. Nagel, "Optical flow estimation: Advances and compar-1097 isons," in Computer Vision ECCV '94, ser. Lecture Notes in Computer 1098 Science, J.-O. Eklundh, Ed. Springer Berlin / Heidelberg, 1994, vol. 1099 800, pp. 49-60. 1100
- [63] B. Heyde, R. Jasaityte, D. Barbosa, V. Robesyn, S. Bouchez, P. Wouters, 1101 F. Maes, P. Claus, and J. D'hooge, "Elastic image registration versus 1102 speckle tracking for 2-d myocardial motion estimation: A direct com-1103 parison in vivo," Medical Imaging, IEEE Transactions on, vol. 32, no. 2, 1104 pp. 449-459, 2013. 1105
- M. D. Cerqueira, N. J. Weissman, V. Dilsizian, A. K. Jacobs, [64] 1106 S. Kaul, W. K. Laskey, D. J. Pennell, J. A. Rumberger, T. Ryan, and M. S. Verani, "Standardized myocardial segmentation and 1107 1108 nomenclature for tomographic imaging of the heart: A statement for 1109 healthcare professionals from the cardiac imaging committee of the 1110 council on clinical cardiology of the american heart association," 1111 Circulation, vol. 105, no. 4, pp. 539-542, 2002. [Online]. Available: 1112 http://circ.ahajournals.org/content/105/4/539.short 1113
- W. Daniel, Applied Nonparametric Statistics. Duxbury Thomson 1114 [65] Learning, 2000. 1115

- [66] V. Chalana and Y. Kim, "A methodology for evaluation of boundary 1116 detection algorithms on medical images," Medical Imaging, IEEE Trans-1117 actions on, vol. 16, no. 5, pp. 642 –652, oct. 1997. D. Liu and E. Ebbini, "Real-time 2-d temperature imaging using 1118
- [67] 1119 ultrasound," Biomedical Engineering, IEEE Transactions on, vol. 57, 1120 no. 1, pp. 12-16, Jan. 2010. 1121
- T. Yingchoncharoen, S. Agarwal, Z. B. Popovi, and T. H. Marwick, [68] 1122 "Normal ranges of left ventricular strain: A meta-analysis," Journal of 1123 the American Society of Echocardiography, vol. 26, no. 2, pp. 185 -191, 2013.
- [69] B. Andres, F. Hamprecht, and C. Garbe, "Selection of local optical flow models by means of residual analysis," in Pattern Recognition, 1127 ser. Lecture Notes in Computer Science. Springer Berlin Heidelberg, 2007, vol. 4713, pp. 72-81.
- [70] M. Pihl and J. Jensen, "Measuring 3d velocity vectors using the 1130 transverse oscillation method," in 2012 IEEE International Ultrasonics Symposium, Dresden, 2012, p. in press.
- M. Pihl, M. Stuart, B. Tomov, J. Hansen, M. Rasmussen, and J. Jensen, [71] "Preliminary examples of 3d vector flow imaging," in SPIE Medical Imaging conference 2013, 2013, p. in press. 1135
- S. Salles, H. Liebgott, D. Garcia, and D. Vray, "Real time 3d us-[72] 1136 tagging combined with 3d phase-based motion estimation." in IEEE 1137 International Ultrasonics Symposium (IUS), Prague, Czech Respublic, 1138 2013, p. in press. 1139

1124 1125 1126

1128 1129

1131 1132